Author's response to reviews

Title: Decision Theory Applied to Image Quality Control in Radiology.

Authors:

Patrícia S Lessa (patlessa@hotmail.com)
Cristofer A Caous (cacaous@gmail.com)
Paula R Arantes (parantes@usp.br)
Edson Amaro Jr (eamaro@usp.br)
Fernando M Campello de Souza (fmcs@hotlink.com.br)

Version: 10 Date: 12 November 2008

Author's response to reviews:

1. The authors list was corrected according to the submission rules.
2. The abstract in the submission system was replaced.
3. The reference list was checked and corrected.